Author's response to reviews

Title: Association between lipids profile and thyroid parameters in euthyroid diabetic subjects: A cross-sectional study

Authors:

Yun ZHANG (wing.1120@163.com)
Ping LU (luping.henan@gmail.com)
Ling ZHANG (zhanglingvip@126.com)
Xinhua XIAO (xiaoxinhua@medmail.com.cn)

Version: 4 Date: 13 January 2015

Author's response to reviews: see over
Dear editor,

I am so apologized for the delay of response. Thank you very much for the informative comments from you and reviewer Altan Onat. We have studied the paper you mentioned and rewritten our Discussion section (Page 8, Paragraph 3-4).

Also, there was a mistake in Table 2. It had been corrected, together with some linguistic errors.

Thanks again to you for the thoughtful and thorough review. Hopefully we have addressed all of your concerns. If there are any problems, please don’t hesitate to tell us.

Best wishes,

Sincerely yours,

ZHANG Yun